## PEER REVIEW HISTORY

BMJ Open publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

## **ARTICLE DETAILS**

## Title (Provisional)

Providing medically tailored groceries and food resource coaching through the charitable food system to patients of a safety-net clinic in Dallas, Texas: a randomized controlled trial protocol

## **Authors**

Albin, Jaclyn; Leonard, Tammy; Wong, Willis; Siler, Milette; Haskins, Carolyn; Turcios, Jessica; Pruitt, Sandi L; Bowen, Michael; Pezzia, Carla; Ford, Alisha; Schinzer, Bianca; Hollis-Hansen, Kelseanna

This article was not externally reviewed at BMJ Open. Protocols that have been independently assessed prior to submission to BMJ Open are usually fast-tracked to publication on the grounds that further substantial changes will not be possible. This independent assessment will usually be external, independent review for both a substantial grant award from a non-commercial or government funder and ethics approval.